Author's response to reviews

Title: Transcriptome-scale similarities between mouse and human skeletal muscles with normal and myopathic phenotypes

Authors:

Alvin T. Kho (alvin_kho@hms.harvard.edu)
Peter B. Kang (peter.kang@childrens.harvard.edu)
Isaac S. Kohane (isaac_kohane@harvard.edu)
Louis M. Kunkel (kunkel@enders.tch.harvard.edu)

Version: 3 Date: 23 February 2006

Author's response to reviews: see over
Re: MS #9342559308882744 - Transcriptome-scale similarities between mouse and human skeletal muscles with normal and myopathic phenotypes

February 22, 2006

Dear Tony:

Thank you for your e-mail message of February 22, 2006. We would like to submit a revised manuscript with the formatting changes you suggest, as well as a handful of grammatical changes.

Please let us know if you have any questions. Thank you for publishing our work.

Sincerely,

Alvin Kho

Peter Kang

Peter B. Kang, M.D.        +1-617-355-5086 (voice)
Program in Genomics, Enders 570  +1-617-730-0253 (fax)
Children’s Hospital Boston  peter.kang@childrens.harvard.edu
300 Longwood Avenue
Boston, MA  02115

Corresponding Authors:

Louis M. Kunkel, Ph.D.        +1-617-355-6729 (voice)
Program in Genomics  +1-617-730-0253 (fax)
Children’s Hospital Boston  kunkel@enders.tch.harvard.edu

Isaac S. Kohane, M.D., Ph.D. +1-617-919-2182 (voice)
Children’s Hospital Informatics Program +1-617-730-0921 (fax)
Children’s Hospital Boston  isaac_kohane@harvard.edu
300 Longwood Avenue
Boston, MA  02115